Author's response to reviews

Title: How useful are post consultation letters to patients?

Authors:

Nicola J Roberts (nicola.roberts@imperial.ac.uk)
Martyn R Partridge (m.partridge@imperial.ac.uk)

Version: 3 Date: 19 December 2005

Author's response to reviews:

Martyn R Partridge
Professor of Respiratory Medicine
Respiratory Medicine
Charing Cross Hospital Campus
St Dunstans Road
London
W6 8RF

19 December 2005

Dear Editor

Thank you for your letter accepting our paper. We have made the formatting changes as requested and are happy for this paper to be published in BMC Medicine instead of BMC Health Services Research.

Kind regards

Martyn R Partridge
Professor of Respiratory Medicine